Author's response to reviews

Title: Mosquito larvicidal activities of Solanum villosum berry extract against the dengue vector Stegomyia aegypti

Authors:

Nandita Chowdhury (cnandita_2006@yahoo.co.in)
Anupam Ghosh (anupamghosh75@yahoo.co.in)
Goutam Chandra (goutamchandra63@yahoo.co.in)

Version: 7 Date: 29 January 2008

Author's response to reviews: see over
Cover letter

To

The Editor,

Sending herewith all the corrections made in the revised manuscript according to the instruction provided by you.

1. Comments: aThese authors contributed equally to this work - please change the a to #

Revision: It has been changed in the revised version of the manuscript

2. Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimize white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here.

Revision: The figure has been cropped according to your instructions and now the figure contains least white space.

From,

Dr. Goutam Chandra.